# Online remote behavioural intervention for tics in 9- to 17-year-olds: the ORBIT RCT with embedded process and economic evaluation

Chris Hollis,<sup>1,2,3,4\*</sup> Charlotte L Hall,<sup>1,2,3</sup> Kareem Khan,<sup>1,3</sup> Marie Le Novere,<sup>5</sup> Louise Marston,<sup>5</sup> Rebecca Jones,<sup>6†</sup> Rachael Hunter,<sup>5</sup> Beverley J Brown,<sup>1</sup> Charlotte Sanderson,<sup>7,8</sup> Per Andrén,<sup>9</sup> Sophie D Bennett,<sup>7,8</sup> Liam R Chamberlain,<sup>1</sup> E Bethan Davies,<sup>1,3</sup> Amber Evans,<sup>7,8</sup> Natalia Kouzoupi,<sup>7,8</sup> Caitlin McKenzie,<sup>1</sup> Isobel Heyman,<sup>7,8</sup> Joseph Kilgariff,<sup>4</sup> Cristine Glazebrook,<sup>1,3</sup> David Mataix-Cols,<sup>9</sup> Eva Serlachius,<sup>10</sup> Elizabeth Murray<sup>5†</sup> and Tara Murphy<sup>7,8</sup>

- <sup>1</sup>NIHR MindTech MedTech Co-operative, Institute of Mental Health, School of Medicine, University of Nottingham, Nottingham, UK
- <sup>2</sup>NIHR Nottingham Biomedical Research Centre, Institute of Mental Health, University of Nottingham, Nottingham, UK
- <sup>3</sup>Mental Health and Clinical Neurosciences, School of Medicine, University of Nottingham, Queen's Medical Centre, Nottingham, UK
- <sup>4</sup>Department of Child and Adolescent Psychiatry, Nottinghamshire Healthcare NHS Foundation Trust, South Block Level E, Queen's Medical Centre, Nottingham, UK
- <sup>5</sup>Research Department of Primary Care and Population Health and Priment CTU, University College London, London, UK
- <sup>6</sup>Division of Psychiatry and Priment CTU, University College London, London, UK
- <sup>7</sup>UCL Great Ormond Street Institute of Child Health (ICH), London, UK/Great Ormond Street Hospital for Children NHS Trust, London, UK
- <sup>8</sup>Psychological and Mental Health Services, Great Ormond Street Hospital for Children NHS Foundation Trust, London, UK

<sup>9</sup>Centre for Psychiatry Research, Department of Clinical Neuroscience, Karolinska Institutet, and Stockholm Health Care Services, Region Stockholm, Stockholm, Sweden <sup>10</sup>Department of Clinical Sciences, Faculty of Medicine, Lund University, Lund, Sweden

\*Corresponding author chris.hollis@nottingham.ac.uk

<sup>†</sup>In memoriam: The ORBIT team would like to acknowledge the contribution made by Elizabeth Murray and Rebecca Jones throughout the project. Both Elizabeth and Rebecca were highly valued colleagues of PRIMENT Clinical Trials Unit, University College London. Rebecca provided a significant contribution to ORBIT, including co-designing the statistical analysis plan and conducting the analysis. Elizabeth provided outstanding expertise in digital trials, which guided the project throughout. We are deeply saddened by their deaths, which occurred during the project.

## **Disclosure of interests**

**Full disclosure of interests:** Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/CPMS3211.

**Primary conflicts of interest:** Chris Hollis was Principal Investigator on a grant from the National Institute of Health Research (NIHR) Health Technology Assessment programme to conduct an Evidence Synthesis on the treatments for tics and Tourette syndrome in children and young people (HTA Project 10/142/01). Rebecca Jones reports NIHR HTA grant funding (NIHR131647). Sophie D Bennett reports the following: grants from NIHR PGfAR programme (RP-PG-0616-20007), Epilepsy Research UK and Beryl Alexander charity; book royalties from *Oxford Guide to Brief and Low Intensity Interventions for Children and Young People*; consultancy fees from Al-Ayn UK (My Story); and income from private practice at Mind and Body London. Isobel Heyman reports personal royalties for several books published with Jessica Kingsley Press on tic disorders. Joseph Kilgariff has received teaching honoraria from the British Association of Psychopharmacology. David Mataix-Cols reports personal fees from UpToDate, Inc., outside the submitted work. Tara Murphy reports royalties from several books published with Jessica Kingsley Press on tic disorders. Elizabeth Murray reports NIHR grant funding (NIHR132243). Cris Glazebrook reports receiving grants from URI and Action for A-T (Charity) and NIHR PGfAR programme (RP-PG-0618-20003). All authors received funding from NIHR to support their salaries and/or time as co-applicants during the conduct of the study.

Published October 2023 DOI: 10.3310/CPMS3211

## **Plain language summary**

Online remote behavioural intervention for tics in 9- to 17-year-olds: the ORBIT RCT with embedded process and economic evaluation

Health Technology Assessment 2023; Vol. 27: No. 18 DOI: 10.3310/CPMS3211

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain language summary**

t can be difficult for children and young people with tics to access therapy. This is because there are not enough trained tic therapists. Online remote behavioural intervention for tics was a clinical trial to see whether an online platform that delivered two different types of interventions could help tics. One intervention focused on techniques to control tics; this type of therapy is called exposure and response prevention. The other intervention was psychoeducation, where participants learned about the nature of tics but not how to control them. The online remote behavioural intervention for tics interventions also involved help from a therapist and support from a parent.

Participants were aged 9–17 years with Tourette syndrome/chronic tic disorder and were recruited from 16 clinics, two study sites (Nottingham and London) or via online self-referral. All individuals who were eligible for the online remote behavioural intervention for tics trial were randomised in a 50/50 split by researchers who were unaware of which treatment was being given. Participants received either 10 weeks of online exposure and response prevention or 10 weeks of online psychoeducation.

A total of 224 children and young people participated: 112 allocated to exposure and response prevention and 112 to psychoeducation. Tics decreased more in the exposure and response prevention group (16% reduction) than in the psychoeducation group (6% reduction) 3 months after treatment. This difference is considered a clinically important difference in tic reduction. The treatment continued to have a positive effect on tic symptoms at 6, 12 and 18 months, showing that the effects are durable. This was achieved with minimal therapist involvement. The cost of online exposure and response prevention to treat young people with tics within this study was less when compared to the cost of face-to-face therapy.

The results show that exposure and response prevention is an effective behavioural therapy for tics in this specific patient group. Delivering exposure and response prevention online with minimal therapist contact can be a successful and cost-effective treatment to improve access to behavioural therapy.

## **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.6

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 3.6 and is ranked 32nd (out of 105 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2021 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), Embase (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### **HTA programme**

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

#### **This report**

The research reported in this issue of the journal was funded by the HTA programme as project number 16/19/02. The contractual start date was in October 2017. The draft report began editorial review in July 2022 and was accepted for publication in December 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2023 Hollis *et al.* This work was produced by Hollis *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

### NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

### **NIHR Journals Library Editors**

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

**Dr Peter Davidson** Interim Chair of HTA and EME Editorial Board, Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Dr Rob Riemsma** Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Helen Roberts** Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk